Author's response to reviews

Title: Assessment of intrahepatic blood flow by Doppler ultrasonography: relationship between the hepatic vein, portal vein, hepatic artery and portal pressure measured intraoperatively in patients with portal hypertension

Authors:

Li Zhang (lilyzhang319_20@hotmail.com)
Jikai Yin (yikfmmuu@hotmail.com)
Yunyou Duan (duanyy@fmmu.edu.cn)
Yilin Yang (yangyl66@126.com)
Lijun Yuan (yuanlj@fmmu.edu.cn)
Tiesheng Cao (caots@fmmu.edu.cn)

Version: 10 Date: 18 July 2011

Author's response to reviews:

Dear editor,

In the revised article, we have revised the authors' name according to your suggest, such as Ji-Kai Yin to Jikai Yin and so on. The reference 9 was added in the end of line 5 at page 6. "All authors read and approved the final manuscript." was added in Author Contribution. And we also rearranged the table and hoped it will be in accordance with the requirements. I'm very appreciated for what you have done for me.

Sincerely
Li Zhang